Author's response to reviews

Title: Evaluation of PROforma as a language for implementing medical guidelines in a practical context

Authors:

David R Sutton (dsutton@brookes.ac.uk)
Paul Taylor (p.taylor@chime.ucl.ac.uk)
Kenneth Earle (Ken.Earle@stgeorges.nhs.uk)

Version: 5 Date: 28 March 2006

Author's response to reviews: see over
Revisions

As requested by the editors we have removed the full stop from the title, ensured that the country is included in all affiliation details, designated Dr. Sutton as the corresponding author, included all authors in references included in the references section, adjusted the figure legends, and cropped the figures.